Author’s response to reviews

Title: Medicinal Cannabis for Psychiatric Disorders: A Clinically-Focused Systematic Review

Authors:

Jerome Sarris (j.sarris@westernsydney.edu.au)
Justin Sinclair (j.sinclair@westernsydney.edu.au)
Diana Karamacoska (d.karamacoska@westernsydney.edu.au; dk744@uowmail.edu.au)
Maggie Davidson (ma.davidson@westernsydney.edu.au)
Joseph Firth (j.firth@westernsydney.edu.au)

Version: 5 Date: 09 Dec 2019

Author’s response to reviews:

1. Please note, the "Availability of Data and Materials" declaration refers to the raw data used in your study.

We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section.

If you do not wish to share your data, please clearly state this in this section along with a justification. For more information and a list of suitable availability statements, please see our Submission guidelines: https://protect-au.mimecast.com/s-/YhcCK1qwBSXMYE51Mn_GH?domain=biomedcentral.com

The manuscript presents a narrative review (rather than meta-analysis) and there is no data available for this study. We have now adjusted this section to state:

“No data or materials were generated for this narrative review.”
2. The specific individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Please add authors Justin Sinclair, Maggie Davidson, and Joseph Dannie Firth. The guidance and criteria for authorship can be found here: https://protect-au.mimecast.com/s/kOA7CL7rxDsGwginUqgzxw?domain=biomedcentral.com

Please represent authors' names using their full initials.

Each author has now been explicitly included in the author contribution section. See below:

Authors’ contributions

The literature search was performed by JS1 and DK. JS1, JS2, DK, MD, and JF contributed to the review of data and preparation of the manuscript. Each author read and approved the final manuscript prior to submission.

3. Since authors (Jerome Sarris and Justin Sinclair) have the same initials, please distinguish between them by designating them with numbers 1 and 2. For example: JS1 and JS2. JS1 would correspond to the author furthest up on the author list.

The authors Jerome Sarris and Justin Sinclair had previously been distinguished as JSa and JSi, as was requested by the editorial team in the previous revision. This has now been changed to the newly requested format of JS1 and JS2, reflecting author order.

Thanks are extended to the editorial team for their assistance with the revision process.